Author's response to reviews

Title: Psychometric Properties of the Chinese Version of the Fatigue Scale-Adolescent

Authors:

William H.C. Li (william3@hku.hk)
Ka Yan Ho (eva19841226@yahoo.com.hk)
Ka Wai Katherine Lam (kathins@gmail.com)
S.Y. Chui (dieselfrankie@yahoo.com.hk)
Chi-Fung Godfrey Chan (devilbb@hku.hk)

Version: 7  Date: 12 November 2015

Author's response to reviews: see over
Dear Editor and reviewers,

Thanks for your comments on my manuscript (MS: 1378078626165418) entitled “Psychometric Properties of the Chinese Version of the Fatigue Scale–Adolescent” I have made changes accordingly, which are highlighted in Red in the text. Please also refer to the following point-by-point responses to the comments from all reviewers and editor. Also, let me know in case any responses are not clear or the information is not adequate to clarify concerns. Thanks for giving us an opportunity to revise the manuscript.

Sincerely,

Authors
### Editorial comments

<table>
<thead>
<tr>
<th>Comment 1</th>
<th>Please clarify in the methods whether all of the participants in this study required parental consent to participate or if some were old enough (17 or 18 year olds for example) to give their own informed consent.</th>
</tr>
</thead>
<tbody>
<tr>
<td>Response 1</td>
<td>Thanks for your question. According to the Hong Kong Institutional Review Board regulations, participant aged 18 years and under requires consent from a parent or guardian. Therefore, all participants in this study required parental consent. We have clarified this in the Methods section.</td>
</tr>
<tr>
<td>Comment 2</td>
<td>Please replace the names of the authors with their initials in the Authors’ contributions section.</td>
</tr>
<tr>
<td>Response 2</td>
<td>We have replaced the names of the authors with their initials in the Authors’ contributions section, thanks!</td>
</tr>
</tbody>
</table>